Author’s response to reviews

Title: Emotion regulation and its relation to symptoms of anxiety and depression in children aged 8- 12 years. Does parental gender play a differentiating role?

Authors:

Mona Elisabeth Lovaas (meslovaas@gmail.com)
Anne Mari Sund (Anne.m.sund@ntnu.no)
Joshua Patras (joshua.patras@uio.no)
Kristin Martinsen (Kristin.martinsen@r-bup.no)
Odin Hjemdal (odin.hjemdal@ntnu.no)
Simon-Peter Neumer (simon-peter.neumer@r-bup.no)
Solveig Holen (solveig.holen@r-bup.no)
Trude Reinfjell (trude.reinfjell@ntnu.no)

Version: 1 Date: 09 Nov 2017

Author’s response to reviews:

Dear Matthew Hickery / BMC Psychology.

Thank you for considering the manuscript for review.

The requested corrections to the manuscript are performed.

Best wishes

Mona E S Løvaas, corresponding author